Author's response to reviews

Title: The puzzle of self-reported weight gain in a month of fasting (Ramadan) among a cohort of Saudi families in Jeddah, Western Saudi Arabia. Balkees Abed Bakhotmah * *Assistant Professor of Nutrition & Food Sciences, Department of Nutrition & Food Sciences, College of Home Economy, Researcher, MHA Chair for DF Research, King Abdulaziz University, P. O. Box 53100 Jeddah 21583,, Saudi Arabia e-mail: hbaab1961@yahoo.com * Corresponding author: Balkees Abed Bakhotmah

Authors:

Balkees A Bakhotmah (hbaab1961@yahoo.com)

Version: 2 Date: 27 May 2011

Author's response to reviews:

Dear Editor,

I have already amended the formatting according to the "Instruction for authors" and your email. Title page, abstract, conclusions were added to the text as requested.